Author's response to reviews

Title: From conceptual pluralism to practical agreement on policy: global responsibility and the case for a Global Fund for Health

Authors:

- Jennifer Prah Ruger (jenpr@upenn.edu)
- Rachel Hammonds (rhammonds@itg.be)
- Gorik Ooms (gooms@itg.be)
- Donna Barry (dbarry@americanprogress.org)
- Audrey Chapman (achapman@uchc.edu)
- Wim Van Damme (wvdamme@itg.be)

Version: 4
Date: 18 August 2015

Author's response to reviews: see over
August 18, 2015

BioMed Central

Dear Mr. Diorelle Gato,

Thank you for the opportunity to revise and resubmit our manuscript (MS: 1131651731133614) entitled, “From conceptual pluralism to practical agreement on policy: global responsibility and the case for a Global Fund for Health,” for further consideration. As requested, herewith we submit our revised manuscript and this cover letter with responses to the referees’ comments. The editorial comments have been incorporated into the final manuscript. Please do not hesitate to contact me if you require further information.

Sincerely,

Jennifer Prah Ruger
Response to Referees’ Comments
MS: 1131651731133614
“From conceptual pluralism to practical agreement on policy: global responsibility and the case for a Global Fund for Health” for BioMed Central

**Referee #1:**

**General comments:**

In general, my comments on the earlier draft are acknowledged and some edits were made. Importantly, some effort has been made to link conceptual issues to practical realities. The article is more of an editorial or opinion piece rather than a journal or research article, but if this is consistent with the journal's policies, I support its publication.

Thank you again for your helpful comments on an earlier draft of our manuscript. We are grateful for your support and kind words.

**Referee #2:**

I am happy with the revisions and the comments. It makes for an interesting read and I suspect the debate that follows will be the most valuable thing.

Thank you again for your interest and kind words. We are grateful for your support and suggestions throughout the revision process.